Author’s response to reviews

Title: How does healthcare worker handwashing behaviour impact upon the transmission of MRSA between patients? : an analysis using a Monte Carlo model

Authors:

Clive B Beggs (C.B.Beggs@bradford.ac.uk)
Simon J Shepherd (S.J.Sheard@bradford.ac.uk)
Kevin G Kerr (kevin.kerr@hdft.nhs.uk)

Version: 3 Date: 16 March 2009

Author’s response to reviews:

1. We have changed this sentence to "From Figure 4 it can be seen that for the worst case scenario (i.e. for a HCW travelling from Patient A [the index case] to Patient B, without washing their hands), the risk of MRSA transmission is approximately 4%.”

2. All remaining references to "handwashing" and "washing their hands" have been replaced with "hand hygiene".

3. The remaining instance of "staphylococcal" has been replaced with "MRSA"